Institut for Matematik og Datalogi Syddansk Universitet

May 2, 2012 JFB DM830 Network Security – F12 – Lecture 7

Lecture, May 2

We will finish chapter 5 and began on chapter 7, introducing PGP.

Lecture, May 9

We will finish chapter 7.

Lecture, May 14

We will begin on chapter 8.

Discussion section: May 16

There will be a presentation on intruders, the first part of chapter 9. Do the following problems:

1. Problem 7.1 in the textbook. 2. Problem 7.2 in the textbook.

3. Problem 7.4 in the textbook (“translated” should be “sent”). 4. Problem 7.5 in the textbook. 5. Problem 7.6 in the textbook. 6. Problem 7.8 in the textbook.

7. Problem 7.9 in the textbook (assume that “zero parity” means that the parity bit is set to zero).